Answering reviewers

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 31046

Manuscript Type: Observational Study

Upper gastrointestinal cancer burden in Hebei Province, China: a population-based study

Li DJ et al. Estimate of Upper gastrointestinal cancer burden

Dao-Juan Li, Di Liang, Guo-Hui Song, Yong-Wei Li, Deng-Gui Wen, Jing Jin, Yu-Tong He

**Dao-Juan Li, Di Liang, Deng-Gui Wen, Jing Jin, Yu-Tong He**, Cancer Institute, the Fourth Hospital of Hebei Medical University/the Tumor Hospital of Hebei Province, Shijiazhuang050011, Hebei Province, China.

**Guo-Hui Song**, Cixian Cancer Institute, Cixian 056500, Hebei Province, China.

**Yong-Wei Li**, Shexian Cancer Institute, Shexian 056400, Hebei Province, China.

Reviewers’ suggestions:
1. Audio core tip:
   In order to attract readers to read your full-text article, we request that the author make an audio file describing your final core tip, it is necessary for final acceptance. Please refer to Instruction to authors on our website or attached Format for detailed information.
2. Please put the punctuation behind reference. Please check throughout. Thank you!
3. Please add PubMed citation numbers and DOI citation to the reference list and list all authors. Please revise throughout. For those references that have not been indexed by PubMed, a printed copy of the first page of the full reference should be submitted.
Answering reviewers:

1. Audio core tip
   Thanks for reviewers’ suggestion. I’ve made the audio core tip.

2. Please put the punctuation behind reference. Please check throughout. Thank you!
   Thanks for reviewers’ suggestion. I’ve check throughout and put the punctuation behind reference of all.

3. Please add PubMed citation numbers and DOI citation to the reference list and list all authors. Please revise throughout. For those references that have not been indexed by PubMed, a printed copy of the first page of the full reference should be submitted.
   Thanks for reviewers’ suggestion. I’ve added PubMed citation numbers and DOI citation to the reference list and list all authors. For those references that have not been indexed by PubMed, a printed copy of the first page of the full reference had been take photos and will be submitted.